Author’s response to reviews

Title: Effect of Prenatal EPA and DHA on Maternal and Cord Blood Insulin Sensitivity: A secondary analysis of the mothers, omega 3, and mental health study

Authors:

Joey England (joey.a.england@uth.tmc.edu)

Joses Jain (joses@pinonperinatal.com)

Bradley Holbrook (Bradholbrook@gmail.com)

Ronald Schrader (RSchrader@salud.unm.edu)

Clifford Qualls (CQualls@salud.unm.edu)

Ellen Mozurkewich (EMozurkewich@salud.unm.edu)

Version: 1 Date: 18 Sep 2019

Author’s response to reviews:

Page 8: specified type of regression

Page 9: now includes sample size for each category, removed row that refers to non-whites

Table 1 first line: changed first line to "showed no significant difference"

Table 2: revised and units changed

Page 10: specified the type of regression analysis used

Page 11: first line is specified that this is baseline measurement

Page 12: specified the type of regression analysis used

Page 4: added subheading introduction

Page 6: replaced words patients and subjects to participants

Page 6: comments on the characteristics of the method used for leptin and adiponectin assays

Page 7: replace word subject to participants